\documentstyle[12pt,fuzz]{article}
\setlength{\topmargin}{0in}
\setlength{\textwidth}{6in}
\setlength{\textheight}{8in}
\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}

\begin{document}

\begin{flushright}
Your Name \\          	% fill in name
Models of Software Systems\\
Homework 1 \\	        % fill in number
Date 			% fill in date
\end{flushright}

\section*{Homework 5: Introduction to Z}		% fill in title

\subsection*{Problem 1}

Solution to Problem 1.

\subsection*{Problem 2}

Solution to Problem 2


\end{document}
